Author's response to reviews

Title: Sexual and reproductive health services for women with disability: a qualitative study with service providers in the Philippines

Authors:

Kira Lee (kira.yy.lee@gmail.com)
Alexandra Devine (devinea@unimelb.edu.au)
Ma. Jesusa Marco (ma.jesusa.marco@dlsu.edu.ph)
Jerome Zayas (jeromebzayas@gmail.com)
Liz Gill-Atkinson (liz.gill-atkinson@unimelb.edu.au)
Cathy Vaughan (c.vaughan@unimelb.edu.au)

Version: 5 Date: 20 September 2015

Author's response to reviews: see over
21st September 2015

Dear Professor Fisher,

RE: Final submission of MS: 8450079181759546 - Sexual and reproductive health services for women with disability: a qualitative study with service providers in the Philippines

We are pleased to hear that this manuscript has been assessed as suitable for publication in the journal. We have made the final editorial revisions requested, as described below.

Editorial requirements:

1. Please provide a Competing Interests section after the List of Abbreviations, according to our Instructions for Authors.

   This oversight has been amended and a Competing Interests section added to lines 653-654.

2. In your Ethics approval section, please specify whether informed consent was verbal or written.

   We have clarified that the informed consent obtained from service providers was written; please see line 214.

We appreciate your consideration of our manuscript, and look forward to hearing from you.

Yours sincerely,

Dr Cathy Vaughan
Centre for Health Equity
Melbourne School of Population and Global Health
The University of Melbourne
Email: c.vaughan@unimelb.edu.au